Author’s response to reviews

Title: Zhenqing Recipe attenuates non-alcoholic fatty liver disease by regulating the SIK1/CRTC2 signaling in experimental diabetic rats

Authors:

Daofei Song (871993733@qq.com)
Lei Yin (m201776027@hust.edu.cn)
Chang Wang (448329145@qq.com)
XiuYing Wen (345144591@qq.com)

Version: 2 Date: 03 May 2019

Author’s response to reviews:

Response: Thank you for your careful review and valuable suggestions again. According to your suggestion, we have applied duplication checking software iThenticate to check duplication rate of our manuscript, and we have reformulated the sections of text in question in the corresponding sections of the revised version, respectively.